Author’s response to reviews

Title: Multiple and Multi-dimensional life transitions in the context of life-limiting health conditions: Longitudinal study focussing on perspectives of Young Adults, Families and Professionals

Authors:

Divya Jindal-Snape (d.jindalsnape@dundee.ac.uk)

Bridget Johnston (Bridget.Johnston@glasgow.ac.uk)

Jan Pringle (Jan.Pringle@ed.ac.uk)

Timothy Kelly (t.b.kelly@dundee.ac.uk)

Rosalind Scott (rosc.scott@btinternet.com)

Libby Gold (libbygold@chas.org.uk)

Raymond Dempsey (therapsgla@ntlworld.com)

Version: 2 Date: 09 Mar 2019

Author’s response to reviews:

-- We note that the current submission contains some textual overlap with other previously published works, in particular:

https://www.chas.org.uk/assets/0001/5082/Multiple_and_Multi-Dimensional_Transitions-Final_Report.pdf

This overlap mainly exists in the methods and limitations sections. PARAPHRASED THROUGHOUT

While we understand that you may wish to express some of the same ideas contained in this publication, please be aware that we cannot condone the use of text from previously published work.

If there is overlap in the Methods section, please ensure to summarize the methods and cite the source. NOT REQUIRED AS PARAPHRASED

Please re-phrase these sections to minimise overlap.

-- Please clarify whether any minors (16+) were included in the patient sample in the methods section of your manuscript. If so, please clarify in the 'Ethics approval and consent to participate'
section whether consent was obtained from a parent or guardian on behalf of any participants under the age of 16. ALL PARTICIPANTS WERE 17 PLUS. A LINE HAS BEEN ADDED TO INDICATE THIS UNDER ETHICS

-- We note that you have included a ‘Consent for publication’ section in the Declarations. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section. DONE

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. DONE